Author's response to reviews

Title: Using Intervention Mapping to develop a programme to prevent sexually transmittable infections, including HIV, among heterosexual migrant men.

Authors:

Mireille E.G. Wolfers (wolfersm@ggd.rotterdam.nl)
Caty van den Hoek (vandenhoekk@ggd.rotterdam.nl)
Johannes Brug (j.brug@erasusmc.nl)
Onno de Zwart (dezwarto@ggd.rotterdam.nl)

Version: 6 Date: 2 July 2007

Author's response to reviews:

Dear editor,

I am delighted to hear you have accepted our manuscript entitled "Using Intervention Mapping to develop a programme to prevent sexually transmittable infections and HIV among heterosexual migrant men." by Mireille E.G.Wolfers et al. (MS: 2034236425111032). According to your instructions I have made some corrections, and uploaded a revised manuscript.

The following changes are made:

Titles are identical now in manuscript and submission system

Unnecessary capitals are removed from section headings

reference to website is made according to your instructions

Furthermore, some minor changes in the text are made. (abbreviations corrected, and reference no.40 in the English title a space was missing)

Yours sincerely,

Mireille Wolfers